Author's response to reviews

Title: We should not be complacent about our population-based public health response to the first influenza pandemic of the 21st century

Authors:

Heath KELLY (heath.kelly@mh.org.au)
Patricia PRIEST (patricia.priest@otago.ac.nz)
Geoff MERCER (geoff.mercer@anu.edu.au)
Gary DOWSE (gary.dowse@health.wa.gov.au)

Version: 6 Date: 2 February 2011

Author's response to reviews:

Dear editorial support team

We have made all the revisions requested in your previous email.
References 1 and 42: please provide functioning web links to the websites. DONE, plus all other links checked and fixed.

Title page: please display the authors' names, affiliations and email addresses in separate sections on the title page. DONE

Author list: please amend the capitalisation of your author list. The preferred style is ‘First name middle Initial Last name’ (e.g. Joe F Bloggs) DONE

Typography: Text re-read and minor errors corrected